Author's response to reviews

Title: Understanding the modifiable health systems barriers to hypertension management in Malaysia: a multi-method health systems appraisal approach

Authors:

Isabelle Risso-Gill (isabelle.risso-gill@lshtm.ac.uk)
Dina Balabanova (dina.balabanova@lshtm.ac.uk)
Fadhlina Majid (fadhlina.majid@gmail.com)
Kien Keat Ng (drkkgng@hotmail.com)
Khalid Yusoff (k.yusoff@salam.uitm.edu.my)
Feisul Mustapha (dr.feisul@moh.gov.my)
Charlotte Kuhlbrandt (Charlotte.Kuhlbrandt@lshtm.ac.uk)
Robby Nieuwlaat (Robby.Nieuwlaat@phri.ca)
J-D Schwalm (schwalm@mcmaster.ca)
Tara McCready (tara.McCreay@phri.ca)
Teo Koon (Koon.Teo@phri.ca)
Salim Yusuf (Salim.Yusuf@phri.ca)
Martin McKee (martin.mckee@lshtm.ac.uk)

Version: 6    Date: 7 June 2015

Author's response to reviews:

As requested, we have changed "Introduction" to Background" in the abstract and main text. We have also added an additional reference. In the text we had referred to related work in Colombia but, when our manuscript was originally submitted, the Colombian paper was only at an early stage of preparation. In the lengthy intervening period it has now been published in its final form so we have cited it. This has not required any change to the text.